Cognostics and Knowledge Used With Dynamical Processing

Claus-Peter Rückemann

Westfälische Wilhelms-Universität Münster (WWU), Leibniz Universität Hannover, North-German Supercomputing Alliance (HLRN), Germany Email: ruckema@uni-muenster.de

Abstract—This paper presents the extended research conducted for an implementation creating knowledge-supported dynamical visualisation and computation, which can also be used in combination with dynamical processing. The focus of methodologies, knowledge, and approach is data-centric, especially concentrating on cognostics and knowledge. The methodologies, knowledge resources, data structures, and workflows are very suitable for long-term multi-disciplinary context and creating application components for the use with High End Computing. The core knowledge is based on long-term knowledge resources further developed for several decades and used with many applications scenarios utilising multi-disciplinary and multi-lingual content and context like references, associations, and knowledge container collections. A major goal of the application case studies shown here is creating context-sensitive dynamical program components and algorithms from selected knowledge content. The selections are results of dynamical workflows, which are part of component implementations, e.g., including search processes and result matrix generators. Previous research has shown that long-term knowledge resources are the most important and most valuable component of long-term approaches and solutions. Here, the structures and classifications are used with independent database frameworks and programming interfaces. The results show that the methodological foundations and knowledge resources are very well suitable as long-term core base, as well as for creating dynamical application components, e.g., for visualisation and computation in multi-disciplinary, geoscientific, and spatial context. The knowledge resources can refer to any kind of resources. The overall environment allows to develop and govern extensive content structures and promote their long-term vitality. The long-term knowledge resources and cognostics are an excellent base for supporting dynamical processing sustainable, economic, long-term development of resources and components.

Keywords–Advanced Knowledge Discovery; Universal Decimal Classification; Conceptual Knowledge; Dynamical Visualisation and Computation; Cognostics.

I. INTRODUCTION

Many methods have been created for deploying computing and processing with knowledge discovery and dynamical visualisation. However, conceptual knowledge and cognostics supporting the creation of fully functional application components has not been been considered explicitely. The target in this implementation is multi-disciplinary information with examples from geoscientific and natural sciences referring to universal knowledge and reaching into many disciplines. The secondary data utilised here can also include spatial information. Within the last decades the value of the content, the "value of data", has steadily increased and with this the demand for flexible and efficient discovery processes for creating results from requests on data sources. This paper presents the extended research based on the creation of knowledgesupported dynamical visualisation and computation, the results of which were published and presented at the GEOProcessing conference in February 2015 in Lisbon [1]. Applying the conceptual knowledge and cognostics in this context, this paper is especially focussing on support for data-centric approaches.

This research shows details of the latest case studies and discusses the up-to-date experiences from the implementation of the dynamical components and their integration with knowledge resources' structures and workflows. The case studies especially consider the levels inside application components, which can be created based on universal knowledge resources and cognostics. The studies also discuss the conceptual support for generators and some resulting features.

This paper is organised as follows. Section II introduces with the state-of-the-art and motivation for this research. Sections III and IV summarise the foundations and challenges with cognostic components, Section V introduces the methodological bases, Section VI presents the previous research being a fundament for this work. Sections VII and VIII present the fundamental implementations, esp. integration referring to knowledge and dynamics, discussing the foundations, architecture, framework, integration, and dynamical visualisation and computation. Section IX discusses selected parts of the implementation and resulting components and Section X presents examples of dynamical cognostic processing in context with knowledge and cognostics. These sections show the details of the implementation and the resulting components, from geo sets, computation to index selection and some views from the resulting visualisations. Sections XI to XIII evaluate the main results, potential and summarise the lessons learned, conclusions and future work.

II. STATE-OF-THE-ART AND MOTIVATION

The creation of long-term knowledge resources and utilisation methods is one of the most pressing goals in information science as the masses of data and the loss of knowledge in society are steadily increasing in all areas. Existing projects employ segment-like spectra of disciplines in their focus. Examples are large digital libraries and projects like the Europeana [2] and the World Digital Library (WDL) [3]. For most data, source and resulting data, it is reasonable to assume a data value at least higher than the funding value [4], which motivates for increased efforts.

As existing projects, e.g., which are only concentrating on bibliographic means, do not focus on such integration, use different and mostly isolated classifications and schemes for different areas and specific purposes. For example, there is a small number of general classifications, which are mostly used in library context. Although such classifications are used in many thousands of institutions worldwide this is neither a general use case or application scenario nor a significant share of the overall knowledge. They are missing to provide facilities for arbitrary kinds of objects, e.g., factual data and trans-disciplinary context in information science and natural sciences. Therefore, the handling of knowledge issues at a cognostic level is not reflected by processes for creation of software components. Besides these major gaps in research, shortcomings result regarding data quality, data-centric solutions, required long-term aspects, and functionality of application components.

In contrast to that, the state of the art for documentation of universal, conceptual knowledge is the Universal Decimal Classification (UDC) [5], which is one of the very few classifications providing a universal classification [6], [7]. Besides public interfaces, the implementations of the known application scenarios are not publicly available in common. Anyhow, all known scenarios have in common that they deploy only a small subset of available classifications and in the vast majority the classification process is not automated. It is necessary to develop logical structures in order to govern the existing big data today and in future, especially in volume, variability, and velocity. This is necessary in order to keep the information addressable and maintain the quality of data on long-term.

Beyond the focus of the mentioned segment-like projects the knowledge resources and concepts discussed and implemented in this research focus on the trans-disciplinary integration of arbitrary different segments and disciplines and a universal usability based on factual data and criteria. The documentation and context also integrates and refers to content and context, e.g., conceptual, procedural, and metacognitive knowledge and allows for a huge range of possible scenarios. This is a driving force to extend the use of classification in trans- and multi-disciplinary context and transfer the experiences from deploying a classification for long-term documentation and application.

III. FOUNDATIONS OF COGNOSTIC COMPONENTS

The term cognostics is related to the Latin verb cognoscere (con 'with', gnoscere 'know, recognise'). This is in itself a cognate of the Greek verb $\gamma \iota \gamma \nu \omega \sigma \kappa \omega$ (gignósko), meaning 'I know, I perceive' (noun: gnósis, 'knowledge') referring to 'recognize', 'conceptualize'. The archaic paradigm of many aspects of fields of human activities is, e.g., a complementary trias of animistics, empirics, and cognostics. Most cognostic content followed prehistorical and early historical times, in the later stages of development. Cognostic fundaments have been discussed in detail by classical ancient Greek philosophers.

"Cognition" as a modern term at least dates back to the 15th century, used for "awareness" and "thinking".

Cognition is the set of all mental abilities and processes. Cognition is therefore related to knowledge, decision making [8], [9], problem solving, human-computer interaction and many more. Cognitive processes use existing knowledge and generate new knowledge.

In result, the implementation of cognostic views can be complex, not only because of application components but especially because of the knowledge-centric base, which has to be created. A lot of following researchers have picked-up the term, e.g., in context with diagnostics guided by computers [10], [11], in an early phase of data exploratory analysis and without developing a base for the knowledge itself.

The term cognostics has been more widely used with developments in information science since the nineteen-nineties. In Geocognostics refers not only to dynamical components but to different views and how to achieve this, e.g., how to integrate political and social-cultural differences, which are examples of major aspects to be considered.

The cognitive aspects of human-computer interaction are multifold, especially in the context of geographic information systems [12]. As one contribution to complex information systems, on the one hand, the accuracy of spatial databases is an important factor [13] but on the other hand it is important that researchers and users can create cognitive collages and spatial mental models [14]. Besides that, the integration of psychological aspects of spatial information [15] as well as the consideration of cultural aspects [16] has become significant for several decades. This has led to the concepts of cognitive geographic information systems [17] and the idea of geocognostics [18]. These fundamental concepts have led to implementations considering cognostics with dynamical features [19] and collaboration frameworks [20], resulting in modular cognostic component concepts [21] and knowledge based approaches and implementations [22], e.g., the new concept of object carousels.

IV. CHALLENGES WITH KNOWLEDGE AND COGNOSTICS

The insufficient care for knowledge has shown to have impact on scientific work [23] and most achievements have to be created over and over again [24] as well as on industrial developments and automation [25], [26], [27], [28], [29], [30].

Data collections can provide any huge amount of information, directly and in consequence of applied workflows. The more unstructured the data collections are the more interpretation of the data are possible. Automated processes can then easily lead to a much higher rate of misinterpretation.

Examples of misinterpretation [31] and unclean data [32], which can regulary result when not integrating high quality knowledge resources have led to suggestions and recommendations that raw data might take enormous efforts to clean, reformat, and consolidate but is very well worth. In spite of the quantity of data [33] bigger does not mean better. For many cases big data has been claimed a 'big misnomer' [34].

The quality of data cannot be neglected especially when creating sustainable structures, long-term documentations and solutions. Some analysis considers data size being important on the one hand, e.g., volume of data. On the other hand, small volumes of data can have a large significance, examples are the following:

- Non-standard mean data like outliers can be significant for working with problems and data.
- Rare data events or attributes can be the interesting ones.
- Rare discrete values or classes.
- Missing values can lead to sparsely observed observation space.

Therefore, the target is a) to minimise the continuous rework and provide long-term features in order to document all parts of the creational steps and results and b) to integrate knowledge and cognostic features, which is a core purpose of the knowledge resources. Besides benefits, complex "all-in" folders, like in election context do show up with challenges [35], which requires alternatives.

V. METHODOLOGICAL BASES: LOGICAL STRUCTURES

To work on that goal requires to define information units and to care for depositing an appropriate segmentation in sub-units. The information units require links to the related units, e.g., superunits. The challenges are to define these structures and units for data used in different disciplinary context, in one discipline, as well as in multi-disciplinary context. These logical structures are the basic precondition for the development of functioning algorithms, which can access the units and whose application can be perfected in a next selflearning step. The tries of using unstructured data result in the fact that data volumes, variabilities, and volumes devaluate the resulting values of requests. Any isolated technological approaches to the big data challenge have shown not to be constructive. A sustainable approach has to consider the data and structure itself.

- The first step is the preconditional definition of a logical, commonly valid structure for the data.
- The second step is the planning for the applications based on the logical structures in step one.
- The third step is the creation of algorithms regarding the data and data retrieval, interfaces, and workflows based on steps one and two. The fourth step is the planning of the implementation. This includes data format, platforms, and applications.

Further, the creation, development, and operation of the content and components require to consider and define the essential plans, especially:

- Plans for extending structures.
- Preparations for all required interfaces for the newly extended structures should be done.
- Plans for self-learning components.
- Plans for container formats and utilisation.
- Plans for sustainable handling of data lifecycles, data formats, and system resources.

The early stage of planning requires a concept catalogue. So far, the activities are commonly in a pre-planning stage. The

next step includes case studies on structures, algorithms, and collaborational issues (efficiency, economical cycle).

VI. FUNDAMENT FOR THIS WORK

Long-term knowledge resources can be created and used for universal documentation and re-use of content. The re-use includes discovery, as well as gathering new results and creating new applications. The knowledge resources [36] can refer to any kind of resources, e.g., to natural sciences resources or historical geographic resources [37]. Basics of knowledge organisation [38] and multi-lingual lexical linked data [39] have been discussed for various disciplines and shown the huge potential and value of the knowledge. This also shows the benefits of linking with universal classifications, especially with consequently numerical notations, which can be easily and most flexibly and efficiently used with modern applications components. Further, on the one hand, information services benefit from a comprehensive and holistic model for evaluation [40] and on the other hand, they align with the benefits for a quality management of information services [41].

The paper presents a new implementation for creating knowledge-supported dynamical visualisation and computation, which have not been integrated before for that purpose. Therefore, a major goal of the application case studies is creating dynamical program components and algorithms based on knowledge resources. The different previous projects and case studies have already shown that the combination of knowledge resources with integrated conceptual knowledge references can be used for the creation of dynamical applications.

The dynamical visualisation and computation based on knowledge resources does have numberless applications. Some prominent examples with the research presented here are knowledge discovery, visualising result matrices from workflows or search processes, and creating objects and extending knowledge resources. The framework presented here is a high level framework interconnecting several frameworks for complex system architecture, multi-column operation, and long-term creation for main resources. Therefore, the required approach is considered to be necessarily most complex from knowledge and implementation point of view.

Following the Geo Exploration and Information case studies [42] based on the actmap framework [43] a number of developments for the deployment of High End Computing resources and technologies with integrated systems are still state of the art. In addition, including the structural and conceptual knowledge based on the knowledge resources, research has been done for a different special database framework, which is as well autonomous and can be used for the creation of standalone dynamical and portable application components. The components can be integrated with the existing frameworks, as well as they can be used as standalone interactive applications.

VII. IMPLEMENTATION: INTEGRATION AND KNOWLEDGE

A. Content and context

The implementation shows the context-sensitive dynamical components based on the knowledge resources. The knowledge resources provide the structure and integrate the factual knowledge, the references, including the references for the object classification views required for the dynamic utilisation, ensuring integration [44] and sustainability [45].

Previous case studies have shown that any suitable cartographic visualisation can be used for the presentation of the results, for example, with the Generic Mapping Tools [46] (e.g., filtering, trend fitting, gridding, views, and projections) or creating exports and imports with various products.

Most available cartographic visualisation products are too specialised in order to handle advanced knowledge workflows on the one hand and dynamical results on cognitive context on the other hand. In the presented case where the application should concentrate on the intention of presenting a special result in an abstract way we require special and flexible facilities for dynamical sketch drawings. The more, in the special case the cognitive background forbids to concentrate on detailed cartographic visualisation or mixing with modern ways of geographic conventions. Historical names, locations, and context are not adequately represented by existing modern frameworks.

Regarding both requirements for this study are fully complied by the flexibility of the implementation. The knowledge resources themselves are not restrictive regarding the use of other components for other purposes.

B. Implementation foundations

The implementation for dynamical visualisation and computation is based on the framework for the architecture for documentation and development of advanced scientific computing and multi-disciplinary knowledge [47]. The architecture implemented for an economical long-term strategy is based on different development blocks. Figure 1 shows the three main columns: Application resources, knowledge resources, and originary resources.



Figure 1. Architecture: Columns of practical dimensions. The knowledge resources are the central component within the long-term architecture.

The central block in the "Collaboration house" framework architecture [48], are the knowledge resources, scientific resources, databases, containers, and documentation (e.g., LX [36], databases, containers, list resources). These can be based on and refer to the originary resources and sources (photos, scientific data, literature).

The knowledge resources are used as a universal component for compute and storage workflows. Application resources and components (Active Source, Active Map, local applications) are implementations for analysing, utilising, and processing data and making the information and knowledge accessible. The related information, all data, and algorithm objects presented are copyright the author of this paper, LX Foundation Scientific Resources [36], all rights reserved. The LX structure and the classification references based on UDC [5], especially mentioning the well structured editions [49] and the multi-lingual features [6], are essential means for the processing workflows and evaluation of the knowledge objects and containers. Both provide strong multi-disciplinary and multi-lingual support.

The three blocks are supported by services' interfaces. The interfaces interact with the physical resources, in the local workspace, in the compute and storage resources the knowledge resources are situated, and in the storage resources for the originary resources.

All of these do allow for advanced scientific computing and data processing, as well as the access of compute and storage resources via services interfaces. The resources' needs depend on the application scenarios to be implemented for user groups.

C. Cognostics and related data

The content-supported cognostics support a plethora of knowledge object sources, information, and data features. An excerpt of examples of these features used in the case studies implemented here are:

- name,
- keywords,
- conceptual knowledge and classification,
- text,
- link references,
- dates,
- languages,
- translations,
- transliterations,
- map locations and GPS,
- various types of comparisons and context references,
- researchers' views,
- comments,
- object specific material, classification, documentation, parameters, files, workflow descriptions, programs, ...

All the data provided for an application, e.g., from knowledge objects in collections or containers, including textual and nontextual data, scientific data, mathematical data, technical data The visualisation components can use all the cognostic attributes in order to further exploit the information for the respective application targets. In addition, with the multidisciplinary long-term features and structures the knowledge resources provide the means for a sustainable integrated research data management [50].

This is significant in combination with integrated realia and digital resources, for example, in digital museums projects [51]. An example for creating components usable with complex environments is an agent-based modeling and simulation in archaeology [52]. This is especially interesting for High End Computing environments, in the given examples, for example, for archaeological simulations [53].

In the presented scenarios implemented with this research, the new resources overcome the insufficient documentation, the structural and long-term deficits. Therefore, the integration benefits from the combination of structured and unstructured data.

A lot of unstructured data is provided on the Internet. For some scenarios this source might provide the largest percentage of unstructured data, regulary integrated in the workflows, e.g., with knowledge discovery processes. Regarding non knowledge resources based sources the propertisation (or 'Proprietarisierung' and 'proprietarisation' in German and French a bit more concisely reflecting the stricter and essential Latin 'proprius', meaning 'own, individual, special, particular, characteristic' and dissociating from 'proper' and 'property') is not only a threat for the free Web [54], [55], [56] but also for the workflows, which can be based on unstructured data and free and open access data, at least to a flexible extend. The situation may lead to challenges with application scenarios, which contain a large percentage of references to unstructured data on the free net as well as with workflows, which require or benefit from such data.

With unstructured Big Data, which is most of the overall amount of data available, there is also a lot of bias [57]. With high end solutions and resources, e.g., with High Performance Computing, there is an increasing demand for techniques supporting fault tolerance [58]. For High End Computing solutions this can be achieved by modular structures and documentation.

One of the important features of algorithms and structures supporting these applications is the ability to cope with failures in information and applied knowledge [59]. With the facilities of the knowledge resources for precision and fuzzyness the components can cover a flexible range of cognostic coverage.

VIII. IMPLEMENTATION: INTEGRATION AND DYNAMICS

A. Integration and computation

The context of the application components is fully integrated with the knowledge resources and dynamical components [60]. The screenshot (Figure 2) illustrates some features. Shown examples illustrate features of Active Source, computed and filtered views, LX information, and aerial site photographs, e.g., from Google Maps.



Figure 2. Dynamical use of information systems and scientific computing with multi-disciplinary and universal knowledge resources [48].

Many general aspects of dynamical use of information systems and scientific computing have been analysed with the collaboration house case studies.

B. Implemented content and application dynamics

The main groups of challenges are resulting from the content and from the applications.

- Content side: From content side, the knowledge resources provide the central repository and infrastructure ('Knowledge as a Service') for discovery and component creation.
- Application side: From application side, the dynamical components can deploy the resources to any extent and in any step of workflows.

Any part of the components and features can be assembled from the knowledge resources' workflows. As an example, the site context and factual data components, the database and the graphical user interface components including event definition and management can be dynamically created via transform routines and concatenate operations.

C. Dynamical visualisation and computation implementation

A number of different visualisation tools and frameworks have been analysed in the latest case studies. The results presented here were mostly realised with Tcl/Tk [61] for the dynamical visualisations, Fortran and C based programs for required algorithms, Message Passing Interface (MPI) [62], and Perl for dynamical scripting.

Many components have been developed for the actmap framework [43] and successfully used and verified in context with existing scenarios. Besides the actmap framework additional possibilities of creating application programming interfaces and graphical user interfaces for dynamical visualisation of knowledge matrices have to be analysed.

As a simple example for a dynamical, portable, and standalone component an application like Tclworld has been considered [63]. The application is built on a very portable Tool Command Language (TCL) base integrating programming, database, and user interfaces.

The database application programming interface [64] is very simple, portable, and extendable. The database graphical user interface [65] can be used within the same application and is based on a rapid prototyping concept. Both interface models allow dynamical control and extension of any features regarding the application, as well as for the content.

IX. IMPLEMENTATION AND RESULTING COMPONENTS

The knowledge resources are very flexibly supporting documentation and handling of knowledge content and context. The features include cognostics as well as dynamical application support.

Key to the flexibility is that any knowledge object can carry as much documentation as required. It is the task of the application workflow to make use of the available information to the extend needed and to develop or create complementary information. Creating knowledge-supported dynamical components starts at the application to knowledge resources' interface level. The conventional components and processes have been described and discussed in practice in detail in previous research, regarding Active Map Software [43], case studies [42], and knowledge integration for classification and computation [44].

For the visualisation and computation using the matrix objects with spatial and georeferenced context, a new application instance "lxworlddynamic" has been created based on the knowledge resources and interfaces. This component re-using the Tclworld interfaces is the required extension complementary to the actmap framework components. All parts of the component shall

- support dynamical and cognostic features,
- integrate with the knowledge resources and existing components, e.g., georeferenced objects,
- be usable interactively,
- have access to the content, e.g., index selection,
- facilitate a standalone application assembly, and
- allow a flexible configuration of all aspects of the applications' visualisation and computation, including cognostic support.

Especially, the component requires a site handler database, has to refer to sets of georeferenced objects, a level handler for selecting levels of detail, to create groups of matrix objects, to support for an individualised configuration, to facilitate index selections on the generated matrix elements, to visualise the matrix elements and context graphically, and has to provide associated data textually and numerically.

Therefore, the implementations done for this study concentrate on creating dynamical, generated, cognostic components. Explicitly, the focus was not on georeferencing standards or cartographic issues or precision. Any of such features can be developed by third parties being interested in supporting their individual scenarios.

A. Knowledge resources and geo sets

The integrated information systems can generate result matrices based on the available components and workflows. The result matrix generators can be configured to deliver any kind of result matrix. One base for the implementation is the generation of georeferences data resulting from requests. The listing (Figure 3) shows an example of a result matrix, an excerpt of the generated site handler database.

1	{LX	Site	20.687652	-88.567674} site 20.687652 88.567674
2	{LX	Site	20.682658	-88.570147} site 20.682658 88.570147
3	{LX	Site	20.682859	-88.568548} site 20.682859 88.568548
4	{LX	Site	21.210859	-86.80352} site 21.210859 86.80352
5	{LX	Site	21.097633	-86.796799} site 21.097633 86.796799
6	{LX	Site	21.157199	-86.834736} site 21.157199 86.834736
7	{LX	Site	21.157199	-86.834736} site 21.157199 86.834736
8	{LX	Site	21.094751	-86.812248} site 21.094751 86.812248
9	{LX	Site	41.377968	2.17804} site 41.377968 -2.17804
0	{LX	Site	41.375842	2.177696} site 41.375842 -2.177696
1	{LX	Site	38.676439	-0.198618} site 38.676439 0.198618
2	{LX	Site	38.677683	-0.198103} site 38.677683 0.198103
3	{LX	Site	21.234502	-86.740494} site 21.234502 86.740494
4	{LX	Site	21.184412	-86.807528} site 21.184412 86.807528
5	{LX	Site	16.043421	-61.663857} site 16.043421 61.663857
6	{LX	Site	16.043153	-61.663374} site 16.043153 61.663374
7	{LX	Site	17.633225	-63.236961} site 17.633225 63.236961
8	{LX	Site	17.633225	-63.236961} site 17.633225 63.236961
9	{LX	Site	51.151786	10.415039} site 51.151786 -10.415039
20	{LX	Site	20.214301	-87.429103} site 20.214301 87.429103
21	{LX	Site	20.493276	-87.735701} site 20.493276 87.735701
22	{LX	Site	20.494663	-87.720294} site 20.494663 87.720294
23	{LX	Site	20.494761	-87.720138} site 20.494761 87.720138
24	{LX	Site	40.821961	14.428868} site 40.821961 -14.428868
25	{LX	Site	20.365228	-87.452545} site 20.365228 87.452545
26	{LX	Site	20.365228	-87.452545} site 20.365228 87.452545

Figure 3. Excerpt of generated site handler database (lxworlddynamic).

The database is the result of a request summarising results on objects referring to a defined context, in this case references between archaeological and geological objects.

The framework provides a number of features like level handlers and sets of object georeferences. The listing (Figure 4) shows an excerpt of the generated site level handler.

```
1 foreach i {
2 { ... }
3 { ... }
4 ...
5 } { + $i level 2}
```

Figure 4. Excerpt of generated site level handler (lxworlddynamic).

The level handler manages the site handler database, which can also be generated and updated dynamically. Appropriate entries are managed by the geo::set. The listing (Figure 5) shows an excerpt of the generated geo set.

```
1 geo::Set {
2 { ... } site ... ...
3 { ... } site ... ...
4 ...
5 }
```

Figure 5. Excerpt of generated geo::set (lxworlddynamic).

Groups of objects, e.g., associated archaeological, geological, meteorite, and volcanological sites, as well as subgroups like pottery and stones, can be dynamically associated and handled in the generated component. The listing (Figure 6) shows an excerpt of the generated database matrix.

```
{Archaeological site} : {A selected site with findings
       of human activity, complementary to a
       {Geological site} .
       These sites have been dynamically created
       from a request to the LX knowledge resources \ldots .
       These sites have been ... .
     {Geological site} : {A selected site with geological
¢
       findings, e.g., a {Volcanological site} or a
10
       {Meteorite site} , complementary to an
11
       {Archaeological site} .
12
13
       This site has been ... }
14
15
      {Metorite site} : {A selected site with meteorite
16
       findings, e.g., meteorite crater, a special
17
       {Geological site}
18
       This site has been ... }
19
20
     {Volcanological site} : {A selected site with
21
       volcanological findings, e.g., volcanological
22
       findings like a volcano or fumarole, a special
23
       {Geological site} .
24
       This site has been ... }
25
                 : {Archaeological site} major.countries {
26
    + Potterv
    Italy France Spain Greece}
27
28
    + Stone
                 : {Geological site} major.countries {Italy
    France Spain Greece}
```

Figure 6. Excerpt of generated database matrix (lxworlddynamic).

Here, the matrix includes site and object types for the respective matrix with excerpts of descriptions and linked references.

B. Index selection and configuration

When a representation of matrix objects in dynamical spatial cartographic context is possible then selected objects can be integrated either from the matrix elements or from the context elements and references (Figure 7).

1	foreach i [geo::Names] {
2	if {[lindex \$geo::db(\$i) 0]=="city"} {
3	"LX_World_database" \$i : {city} loc [lrange \$
	geo::db(\$i) 1 end] }
4	if {[lindex \$geo::db(\$i) 0]=="mount"} {
5	"LX_World_database" \$i : {mount} loc [lrange \$
	geo::db(\$i) 1 end] }
6	if {[lindex \$geo::db(\$i) 0]=="site"} {
7	"LX_World_database" \$i : {site} loc [lrange \$
	geo::db(\$i) 1 end] }
8	if {[lindex \$geo::db(\$i) 0]=="lake"} {
9	"LX_World_database" \$i : {lake} loc [lrange \$
	geo::db(\$i) 1 end] }
10	if {[lindex \$geo::db(\$i) 0]=="road"} {
11	"LX_World_database" \$i : {road} loc [lrange \$
	geo::db(\$i) 1 end] }
12	if {[lindex \$geo::db(\$i) 0]=="rail"} {
13	"LX_World_database" \$i : {rail} loc [lrange \$
	geo::db(\$i) 1 end] }
14	if {[lindex \$geo::db(\$i) 0]=="river"} {
15	"LX_World_database" \$i : {river} loc [lrange \$
	geo::db(\$i) 1 end] }
16	if {[lindex \$geo::db(\$i) 0]=="grid"} {
17	"LX_World_database" \$i : {grid} loc [lrange \$
	geo::db(\$i) 1 end] }
18	}

Figure 7. Excerpt of generated index selection (lxworlddynamic).

The listing shows an excerpt of the generated index selection. Matrix elements are, e.g., sites. Context elements and references are, e.g., cities, mounts, lakes, roads, rails, rivers, and grids. Any part of the dynamically generated components can be individualised depending on context-sensitive attributes and workflow configuration. The listing (Figure 8) shows an example for the generated on-the-fly-symbol used for "sites".

1	<pre>set bitmaps(site) [imag "</pre>	e create bitmap	-data [strimj::xbm
2	2 #		
3	3 # # #		
4	4 # #		
5	5 .#.###.#.		
6	6 ##.###.##		
7	7 .#.###.#.		
8	8 # #		
9	9###		
10	10#"] -foreground	darkviolet]	

Figure 8. Excerpt of generated on-the-fly symbol for sites (lxworlddynamic).

Different symbols can be integrated for different sites or for different groups. The objects with their symbols are only visible if the defined level, which is handled by the level handler, is active in the interactive view.

C. Dynamical visualisation and computation

The following image (Figure 9) shows a screenshot of a resulting dynamical visualisation of items in the result matrix, in this case the resulting archaeological context sites. The generated application utilises all the features so far described with the implementation.



Figure 9. Archaeological context sites in interactive, dynamically generated spatial application (lxworlddynamic).

The screenshot illustrates the dynamical visualisation of the matrix elements for the context of the respective results. With the workflow a spatial context has been chosen for the matrix, creating the components. The spatial application component has been assembled by the workflow, integrating the object item references from the result matrix and secondary information from the referring knowledge resources objects with a dynamical and interactive view of the matrix.

Figure 10 shows a screenshot of a resulting dynamical zoom visualisation of matrix results and secondary information on geological and archaeological context sites. The partially shown superpositioning effect of the respective zoom is still visible in order to show the results, which can be separated in different cognitive views, different zooms, and event sensitive actions. The implementor can do anything with this feature he is interested in, e.g., use the interactive features for label stacks and level effects, being sensitive for single results or result group. Workflow sensitive cartographic material objects (e.g., cities, land, sea, countries, border lines, grid lines) are shown for orientation and context and support cognitive feature display. The shown zoom value and the scroll bars indicate the level of detail. The screenshot illustrates the matrix elements and their references within the knowledge resources.



Figure 10. Zoom archaeological and geological sites and context, integrated in interactive, dynamically generated spatial application (lxworlddynamic).

In this example, an active context-sensitive window component is delivering the secondary information. This component is actively communicating with the other components. The site entity data regarding levels and objects referred from the matrix elements is dynamically available (Figure 11).

1	LX Site	16.043153,-61.663374
2		
3	level	2
4	:	site
5	loc	16.043153 61.663374

Figure 11. Example of a single site entity data extract (lxworlddynamic).

Here the matrix elements are referring to the attributes of the knowledge resources' objects, e.g., sites and cities. In this example, the displayed data excerpt includes the level, the type, and the location of the respective site. Figure 12 shows a screenshot of the corresponding dynamical site database.

💥 LX World database		×
LX Site 16.043153,-61.663374 LX Site 16.043421,-61.663857 LX Site 17.633225,-63.236961 LX Site 20.214301,-87.429103 LX Site 20.365228,-87.452545	⊥ LX Site 16.043153,61.663374 level 2 i site loc 16.043153 61.663374	7
Find in LX World database:	90-07-	

Figure 12. Site database, secondary information (lxworlddynamic).

The site database can be accessed by integrated or external applications' components, e.g., for searches, references or for

X. DYNAMICAL COGNOSTIC PROCESSING

Some of the knowledge object data is processed and filtered for cognostic context and shown in application components, e.g., as secondary data in dynamical windows, e.g., in the LX World database windows.

A. Knowledge levels and cognostics

A resulting selection can, for example, be used for interactive discovery processes in the dynamically generated result matrix. The next passages are based on a result matrix of objects from the context of natural sciences and archaeology. The result matrix elements were dynamically processed and added to the interactive application component. The standard entries had not to be removed as from the processing workflow they are not interlinked and as they demonstrate the coexistence of non-linked objects. The data is based on the knowledge resources and the lxworlddynamic components. The workflow illustrates the secondary information from the dynamically extendable database being used for cognostic based improvement of an intermediate result matrix. In this case, the goal is to look for archaeological sites with associations to more than two societies or cultures as they might be called. The following example (Figure 13) shows a search window searching "Society" context in the intermediate result matrix in the intermediate historical result matrix.

💥 LX World database	
Society: Greek Society: Mayan Society: Roman Sofia Soufriere St Peterburg Stockhom	↓ LX World: Result matrix objects for 'Society:': Society: Mayan Society: ({Society:}) {Arch pos=0 Society: ({Society:}) {Arch pos=0 Society: ({Society:}) {Arch pos=0 Society: ({Society:}) {Arch pos=0 Society: ({Society:}) {Arch
Find in LX World database	: Society:

Figure 13. Secondary information on "Society" (lxworlddynamic).

The secondary search window contains the results from within the result matrix, Mayan, Roman, and Greek society. They are dynamically interlinked with deeper respective information. The following example (Figure 14) shows the referenced information for "Mayan society" in the context of the result matrix.

💥 LX World databas			×
Society: Mayan Society: Roman Sofia Soufriere St Peterburg Stockholm Stoce	△ Society: Mayan Society: Society: Archaeological major.sites Cenote Sagrado Chichen I El Meco IK Kil Isla Mujeres Templo de Al Tulum Xel Ha 7	<u>site</u> <u>tza C</u> Lacran	<u>oba</u>
Find in LX World databas	e: Society:		!

Figure 14. Dynamically linked information on "Mayan society" (lxworlddynamic).

The following example (Figure 15) shows the referenced information for "Greek society" in the result matrix' context.

Society: Greek Society: Mayan Society: Roman Sofia Soufriere St Peterburg Stockholm	△ Society: Greek Society: Society: Archaeological site major.sites Antipolis Athens Delphi Halkidiki Kos Marathon Rhodos Sparta I ✓ I
---	---

Figure 15. Dynamically linked information on "Greek society" (lxworlddynamic).

The following example (Figure 16) shows the referenced information for "Roman society" in the result matrix' context.

Society: Roman Sofia Soufriere St Peterburg Stockholm Stone Svdnay	Society: Roman Society: Society: <u>Archaeolog</u> major.sites <u>Altino Antipolis Ea</u> <u>Cemenelum Valentia Edetanorum</u> ✓	ical site rcino Bastulo
--	--	----------------------------

Figure 16. Dynamically linked information on "Roman society" (lxworlddynamic).

Whereas the first entry does not carry a dynamical reference to a common location, both entries Greek and Roman link to a location object "Antipolis".

The following example (Figure 17) shows the referenced information for "Antipolis" in the context of the result matrix.

Antipolis Archaeological site	Antipolis	
Arizona		
Arkansas	Object: Archaeological site in France,	country in
Asia	Europe	
Asuncion	capital <u>Paris</u>	
Athens	country. language Greek Latin	

Figure 17. Dynamically linked information on "Antipolis" linked as well from "Greek Society" and "Roman society" (lxworlddynamic).

When the discovery goes into the elements' context, then the attributes refer to many details of the context. Figure 18 shows an excerpt of an referenced Antipolis object entry with UDC classified knowledge objects.

1	Antipolis [Archaeology, Geophysics, Remote Sensing]:
2	G	reek city, later Roman city, Southern France.
3	M	Modern location name Antibes, between Nice and
		Cannes, France.
4	P	redecessor of the city of Antibes, France.
5	90	%IML: UDC:711(38)(37)=12=14=13

Figure 18. Content-supported cognostics: Knowledge object "Antipolis" (LX resources).

This not only refers to the antique Greece and to the antique Rome and to both cultural 'areas' but also to a related predecessor name of the city. Table I shows the referred conceptual knowledge.

Table I. Universal Decimal Classification location and language (knowledge resources, excerpt, English version).

UDC Code	Description	
UDC:(37) UDC:(38) UDC:=12 UDC:=13 UDC:=14	Italia. Ancient Rome and Italy Ancient Greece Italic languages Romance languages Greek (Hellenic)	

As used in the objects' excerpt, the classification presents a view, which includes a cognostic meaning with its selection and sort order. In this case, only appropriate elements from the compiled result matrix are linked automatically when relevant for the application scenario.

B. Synonyms and cognostics

The following example (Figure 19) shows a search window searching in the intermediate result matrix for the synonyms of Vesuvius.

💥 LX World database		×
Synonym: Vesuvius Tehran Texas Tianjin Tokyo Toronto Turum	Synonym: Vesuvius Synonym: Synonym: <u>Volcanological site</u> Italy, country in Europe, Synonym name: Mons Vesuvius, Vesuvio, Vesuv, Vesaevus, Vesevus, Vesbius, Vesvius	in
Find in LX World database:	Synonym	1

Figure 19. Dynamically linked information on "Vesuvius" synonyms (lxworlddynamic).

The excerpt result matrix provides quite a number of Vesuvius synonyms. Any further discovery can exploit this context for views and cognostics, either proceeding with the discovery in-width or in-depth.

Figure 20 shows an excerpt of an referenced Vesuvius object entry with UDC classified knowledge objects.

1	Vesuvius	[Vulcanology, Geology, Archaeology]:
2		(lat.) Mons Vesuvius.
3		(ital.) Vesuvio.
4		(deutsch.) Vesuv.
5		Volcano, Gulf of Naples, Italy.
6		Complex volcano (compound volcano).
7		Stratovolcano, large cone (Gran Cono).
8		Volcano Type: Somma volcano,
9		VNUM: 0101-02=,
10		Summit Elevation: 1281\UD{m}.
11		The volcanic activity in the region is observed
		by the Oservatorio
12		Vesuviano. The Vesuvius area has been declared a national park on
13		\isodate{1995}{06}{05}. The most known antique
		settlements at the
14		Vesuvius are Pompeji and Herculaneum.
15		Syn.: Vesaevus, Vesevus, Vesbius, Vesvius
16		s. volcano, super volcano, compound volcano
17		s. also Pompeji, Herculaneum, seismology
18		compare La Soufrière, Mt. Scenery, Soufriere
19		%%IML: UDC:[911.2+55]:[57+930.85]:[902]"63"
		(4+23+24) = 12 = 14
20		<pre>%%IML: GoogleMapsLocation: http://maps.google.de</pre>
		/maps?hl=de≷=de&vpsrc=0&ie=UTF8≪
		=40.821961,14.428868&spn=0.018804,0.028238&t=h&
		z=15

Figure 20. Content-supported cognostics: Knowledge object "Vesuvius" (LX resources).

Besides all the other information in this excerpt, the names of the object in different languages and the explicitly declared synonyms were choosen to be handled equally when dynamically generating the "Synonym:" data for the lxworlddynamic database as shown in Figure 19. Due to the reference to the objects in the knowledge resources it will also be possible to use the information separately again for further cognostic purposes, e.g., when creating a language specific search interface.

It is important to mention that with the excerpts of classification shown the conceptual knowledge consistency is supported by editions, which may be used exclusively but also in combination. Table II excerpts the classification (UDC) references used for objects as implemented with the knowledge resources references for the example above.

Table II. Universal Decimal Classification "Vesuvius" (knowledge resources, excerpt, English version).

UDC Code	Description
UDC:55	Earth Sciences. Geological sciences
UDC:57	Biological sciences in general
UDC:902	History of civilization. Cultural history
UDC:902	Archaeology
UDC:63"	Archaeological, prehistoric, protohistoric periods
UDC:(4)	Europe
UDC:(23)	Above sea level. Surface relief. Above ground
UDC:(24)	Below sea level. Underground. Subterranean
UDC:=12	Italic languages
UDC:=14	Greek (Hellenic)

Whereas any of the application components are limited by default –not only those in the examples above– the resources containing the knowledge provide much higher flexibility, which includes "universal" conceptual knowledge features. In consequence, only individual representations and views of the objects can be shown.

C. Cognostic levels and dynamical visualisation

Classification has been considered as a research tool for several decades now [66]. When getting into non-disciplinary classification [67] or the more into integrated knowledge processing then the UDC can support even complex context. Today, many sites using a classification will considered UDC to be the top end of the existing multi-lingual universal classifications [68]. In detail, regarding some disciplines' classification the reliability of classification is an issue, e.g., with diagnostic classification [69]. In other cases, e.g., with text classification a non-parametric statistical method can help to provide solutions [70]. The recherche in heterogeneous data resources is a special challenge [71], which can be supported by referencing between classifications and imaging between classifications.

This means besides UDC, regarding universal, multidisciplinary knowledge, natural sciences, and mathematics, e.g., the Physics and Astronomy Classification Scheme (PACS) [72], the Mathematics Subject Classification (MSC) [73], the U.S. Geological Survey (USGS) library classification system [74], and the North American Industry Classification System (NAICS) [75] are and were widely in use and might need to be considered.

With the research on knowledge resources and dynamical components basic work for intermediate classification has been done [76], which can extend the knowledge context and interlinks specialised resources into universal conceptual knowledge. Handling details or respectively the content and context of "objects" includes challenging tasks. Using an universal classification can support the cognostic aspects in many ways. For example, on the one hand, the granularity must be as high as possible but on the other hand it should still be easy to group certain selections. Therefore, the documentation requires means, which combine precise and fuzzy methods at the same time. This results in challenges from creating documentation down to using the content in any way. Today advanced cognitive geoscientific information system components can integrate higher levels of cognostic information. All the aspects can be documented and provided applying a universal classification. Mapping cognostic details with levels of details can, e.g., be aligned with the depth of the classification.

All the objects, like map objects and their attributes are part of the environment, including the city markers. These entries are defined for creating a specific environment.

In this example, from the second view level on the generated entries are shown. These entries are objects from the group of result matrix elements. The context and attributes of these objects can be generated according to the cognostic context and purpose. For example, the labels, colours, appearance of the objects, the dynamical features and so on can be modified by the author for creating the respective view accordingly. In the above examples cognostics contributes especially from the

- Object level (e.g., documentation context),
- Workflow level (e.g., result matrix generation),
- Application level (e.g., component implementation),
- Interactive level (e.g., dynamical context).

The following screenshot (Figure 21) shows an example for the dynamical visualisation of a result matrix containing archaeological and geological sites and context. The object data is integrated in an interactive, dynamically generated spatial application. The screenshot shows level 1, where the cognostic level of detail containing the "site" objects is defined level 2.



Figure 21. lxworlddynamic – archaeological and geological sites and context, integrated in interactive, dynamically generated spatial application (level 1).

Whereas in this example level 1 only contains continents, some border lines, and some larger cities level 2 also visualises the targeted objects.

D. Conceptual support for generators

The high level of required application and discovery processes leads to the fact that there must be facilities to create result matrices via many different ways when using a knowledge resources concept.

Structured long-term resources integrating complex documentation and classification and providing the required 'data features' have been developed for that purpose. Besides the large spectrum of features of the knowledge objects, the major attribute supporting cognostic reuse is conceptual knowledge.

The knowledge object can carry many attributes, e.g., references, keywords, and conceptual knowledge. In many cases this conceptual knowledge comes in the form of one or more classifications.

The classification, which has shown up being especially important for complex multi-disciplinary long-term classification with knowledge resources is the Universal Decimal Classification (UDC) [77].

UDC allows an efficient and effective processing of knowledge data and provides facilities to obtain a universal and systematical view on classified objects.

Regarding library applications only, UDC is used by more than 144,000 institutions and 130 countries [78]. Further operational areas are author-side content classifications and museum collections, e.g., with documentation of resources, library content, bibliographic purposes on publications and references, for digital and realia objects.

The tiny unsorted excerpts of knowledge resources objects only refer to main UDC-based classes, which for this part of the publication are taken from the Multilingual UDC Summary (UDCC Publication No. 088) [49] released by the UDC Consortium under the Creative Commons Attribution Share Alike 3.0 license [79] (first release 2009, subsequent update 2012).

If nothing special is mentioned the basic classification codes are used in an unaltered way. If a classification refers to a modified code in particular contributing authors have to notice and document the modifications explicitely. So in practice, the classification views with knowledge resources are references to UDC.

E. Series of cognostic views

The series of screenshots in Figure 22 depicts the sequence zoom details in levels 2 to level 5 for the archaeological and geological sites and context. The first image shows the LX sites generated from the result matrix. The second image shows the zoom of the Europe LX sites' area. The third and fourth images show a zoom and a followup zoom of the Europe selection, e.g., identifying Barcin and Vesuvius sites.



Figure 22. lxworlddynamic – archaeological and geological sites and context, integrated in interactive, dynamically generated spatial application (top to bottom: Zoom details, level 2 - 5).

F. Triggering events and interactive features

Figure 23 shows the generated search window for the result matrix site database, especially with the Vesuvius georeference data.

LX Site 38.676439,-0.198618 LX Site 38.677683,-0.198103	LX Site 40.821961,14.428868	-
LX Site 40.821961,14.428868	Level 2	
LX Site 41.375842,2.177696	loc 40.821961 -14.428868	
LX Site 41.377968.2.17804	7	

Figure 23. lxworlddynamic – archaeological and geological sites and context, site database window (Vesuvius).

Using the "!" function triggers a "goto" for the coordinates and open a corresponding zoom window containing the selected georeferences (Figure 24).



Figure 24. lxworlddynamic – archaeological and geological sites and context, "goto" (!) zoom window (Vesuvius).

The sequence of the consequent levels and interactive features shows a dynamical visualisation integrated with result matrices and components generated from universal knowledge resources. Based on this any events can be triggered that way, e.g., visualisation or computation events. This goto function is only an example for the multitude of possibilities to add code and procedures to the generated components in order to support the dynamic and cognostic features.

G. Generating referenced information code

Figure 25 shows the generated search window for the result matrix site database, especially with a country dependent graphics included with the information. The code for the component as well as for the content has been created dynamically.



Figure 25. lxworlddynamic – archaeological and geological sites and context, site database window (Coba).

The listing (Figure 26) shows an excerpt of the generated database entry code for the database information windows.

1	+ Cimiez @ flags/fr.gif : {{Archaeological site} in France, country in Europe} capital Paris country. code FR
2	+ Altinum @ flags/it.gif : {{Archaeological site}
	in Italy, country in Europe} capital Rome country.code IT
3	+ Barcin @ flags/es.gif : {{Archaeological site}
	in Spain, country in Europe} capital Madrid country. code ES
4	+ Coba @ flags/mx.gif : {{Archaeological site}
	in Mexico, country in America}
5	+ Tulum @ flags/mx.gif : {{Archaeological site}
	in Mexico, country in America}
6	+ Vesuvius @ flags/it.gif : {{Volcanological site}
	in Italy, country in Europe} capital Rome country.code
	IT
7	+ Soufriere @ flags/fr.gif : {{Volcanological site}
	on Guadeloupe, France, Caribbean, F.W.I.}
8	+ {Mt. Scenery} @ flags/nl.gif : {{Volcanological site}
	on Saba, The Netherlands, Caribbean, D.W.I.}

Figure 26. Excerpt of generated database entry code for the database information windows (lxworlddynamic).

The excerpt includes the references for the hypertext and flag code for parts of the result matrix elements. The database code can be used for many purposes, as codes generated like this part are capable of being directly integrated into the dynamical application components as shown above. The text is automatically linked by the framework environment in a very simple fuzzy way and can transport cognostic aspects in textual and visual context. The cognostic levels can be handled very flexible, e.g., the content and the generated database entry code in arbitrary ways. In this case, the result matrix defines objects for one level. The environment defines additional context objects for other levels.

XI. EVALUATION

Application components can be created and assembled dynamically from any workflow. Knowledge objects can be used efficiently with any dynamical components. The cognostic features for the content as well as for resulting application components can be efficiently provided by conceptual knowledge documented with advanced knowledge resources. Database interfaces can be used dynamically and efficiently with the components. Graphical User Interfaces (GUI) and Application Programming Interfaces (API) can be used most flexible on that dynamical base.

The information created from arbitrary numbers of resources' objects in this excerpt includes site labels, level information, category information, as well as the georeferences. The components can trigger any instances and events dynamically and interactively. This allows any kind of processing, computation, and visualisation, from sketch like visualisation to special cartographical mapping. All the more, the dynamical components based on the knowledge resources and the lxworlddynamic frameworks allow for the interactive, autonomous components' generation. The components have been successfully implemented on a number of operating systems (e.g., SuSE Linux, Debian, Red Hat, and Scientific Linux, as well with older and up-to-date distributions). The knowledge and system architecture allows to seamlessly integrate with all the steps required for a sustainable implementation of the methodological bases. The case studies done over several decades of knowledge resources' creation and development and two decades of application component developments have shown that plans for extending structures, creating interfaces, self-learning components, container formats, and integration with life cycles and systems resources' operation can be assured even for long-term application.

XII. FURTHER POTENTIAL OF COMPONENTS

The development with this long-term research has led to sustainable solutions when implementing components that support splitting up the data, which is most important in order to achieve an efficient use of available resources, e.g., Input/Output or distribution onto many disks and machines. Otherwise, bulk data can be as inefficient to handle as it will not be practical to use certain methods and algorithms at all. Not all the tools, which are implemented by respective disciplines for handling data are suitable and efficient for all data, algorithms, and workflows. Parallel processing can and should be done with different tools for different purposes. Examples are commonly used tools like Hadoop, Hadoop Distributed File System (HDFS), and MapReduce with a basic Application Programmers Interface (API) for a more abstracted way of developing components for parallel processing of portions of data on a cluster system.

For critical cases it might be argued that components like MapReduce code are costly to write and difficult to troubleshoot and manage. With the same breath, tools like MapReduce are inefficient for exploring datasets. Therefore, the Hadoop_on_a_chip 'ecosystem' got tools that span a wide variety of needs, including providing layers of abstraction that make interacting with the data simpler.

In public presentation and marketing common understanding of containers is reduced to certain aspects, e.g., security features [80], [81]. Special container concepts have been introduced for handling Big Data, especially scientific data, e.g., the NERSC 'Shifter' at the National Energy Research Scientific Computing Center [82].

These approaches are insufficient from the content and knowledge related point of view of containers. For example, on the one hand, there is nothing general for a container when postulating that it should contain "everything". At the same level, security features have to be considered very special with specific cases of application. In addition, in most cases the advanced application scenarios trigger those secondary conditions. On the other hand, containers must not only provide computational features. The result is that up to now we neither have a commonly discussed container concept nor a set of universal features. A container is a term for a data or file format bundling the data for certain purpose and application. The features for anything more interesting will include data, documentation, references and so on. Doing so includes how the information is transferred or accessed, which includes to define the modalities for a certain scenario. When discussing containers from the content and knowledge view

many scenarios if not universal application is the target, which induces a much more general understanding of containers as has been done with this research.

XIII. CONCLUSIONS

Application components have shown to benefit in many cases from dynamical processing in context with knowledge and cognostics. This research has provided details of the implementation and the resulting components, from structure, content, and context of knowledge resources and cognostic support for dynamical components, geo sets, computation to index selection and various views from the resulting visualisations.

Long-term knowledge resources are core means for providing a sustainable base for these features. The knowledge resources can refer to any kind of resources and allow to efficiently transport implementation features for long-term vitality of use cases. Over many years the implementation with different scenarios has proven to be straight forward.

It has been demonstrated that with the proposed framework and concept context-sensitive dynamical components can be successfully created on base of universal knowledge resources. It has been shown that even standalone dynamical components can be created based on the implementation of the foundations, supporting arbitrary dynamical, modular, portable, and extendable database application programming and database graphical user interfaces. The implementation can utilise workflows and algorithms for knowledge discovery and selection up to intelligent application component creation. It integrates very efficiently in workflow chains, e.g., for computation and visualisation, and is very well usable even for rapid prototyping environments.

The integration is non-invasive regarding the knowledge resources for uni-directional visualisation and computation. If the intention with an application scenario is to update information consistently then multi-directional workflows can also update objects in side knowledge resources or containers from the created components in arbitrary ways, ensuring consistency and plausibility, as well as following management and security policies. Future work will be focussed on issues and usability beyond the plain Big Data approaches, resulting from data values, creating implementations supporting data vitality. A major focus will concentrate on sustainable integration of content and context with conceptual knowledge and cognostics.

ACKNOWLEDGEMENTS

We are grateful to all national and international partners in the GEXI cooperations for the innovative constructive work. We thank the Science and High Performance Supercomputing Centre (SHPSC) for long-term support of collaborative research since 1997, including the GEXI developments and case studies. We are grateful to the "Knowledge in Motion" (KiM) long-term project and its scientific members, DIMF, for the contributions on the development and application of classifications, especially to Dr. Friedrich Hülsmann, Gottfried Wilhelm Leibniz Bibliothek (GWLB) Hannover, to Dipl.-Biol. Birgit Gersbeck-Schierholz, Leibniz Universität Hannover, and to Dipl.-Ing. Martin Hofmeister, Hannover, for practical multidisciplinary case studies and the analysis of advanced concepts. Many thanks go to the Institute for Legal Informatics (IRI), Leibniz Universität Hannover, and to the scientific colleagues at the Westfälische Wilhelms-Universität (WWU), for discussion, support, and sharing experiences on collaborative computing and knowledge resources and for participating in fruitful case studies, as well as to the participants of the postgraduate European Legal Informatics Study Programme (EULISP) for prolific discussion of scientific, legal, and technical aspects over the last years. We are grateful to Dipl.-Ing. Hans-Günther Müller, Cray, for previous work on flexible practical solutions to architectural challenges and excellent technical support. Last but not least we thank Richard Suchenwirth for his initial implementation of Tclworld, which could be used as a tool for handling result matrices.

REFERENCES

- C.-P. Rückemann, "Creating Knowledge-based Dynamical Visualisation and Computation," in Proceedings of The Seventh International Conference on Advanced Geographic Information Systems, Applications, and Services (GEOProcessing 2015), February 22 – 27, 2015, Lisbon, Portugal. XPS Press, 2015, pages 56–62, ISSN: 2308-393X, ISBN-13: 978-1-61208-383-4, ISBN-13: 978-1-61208-048-2 (CDROM), URL: http://www.thinkmind.org/download.php?articleid= geoprocessing_2015_3_40_30063 [accessed: 2015-04-12].
- [2] "Europeana," 2015, URL: http://www.europeana.eu/ [accessed: 2015-04-12].
- [3] "WDL, World Digital Library," 2015, URL: http://www.wdl.org [accessed: 2015-04-12].
- [4] F. Hülsmann and C.-P. Rückemann, "Funding Value and Data Value," KiM Summit, June 15, 2015, Knowledge in Motion, "Unabhängiges Deutsches Institut für Multi-disziplinäre Forschung (DIMF)", Hannover, Germany, 2015.
- [5] UDC, Universal Decimal Classification. British Standards Institute (BSI), 2005, complete Edition, ISBN: 0-580-45482-7, Vol. 1 and 2.
- [6] "UDC Online," 2015, http://www.udc-hub.com/ [accessed: 2015-04-12].
- [7] "Universelle Dezimalklassifikation (UDK)," 2014, Uni-Protokolle, URL: http://www.uni-protokolle.de/Lexikon/Universelle_ Dezimalklassifikation.html [accessed: 2015-04-12].
- [8] C.-P. Rückemann, "Decision Making and High End Computing," International Tutorial, October 21, 2012, The Second International Conference on Advanced Communications and Computation (IN-FOCOMP 2012), International Conference on Advances in Information Mining and Management (IMMM 2012), International Conference on Social Eco-Informatics (SOTICS 2012), International Conference on Mobile Services, Resources, and Users (MOBIL-ITY 2012), International Conference on Global Health Challenges (GLOBAL HEALTH 2012), International Conference on Communications, Computation, Networks and Technologies (INNOV 2012) / DataSys 2012, October 21–26, 2012, Venice, Italy, [Lecture], 2012, URL: http://www.iaria.org/conferences2012/filesINFOCOMP12/ rueckemann_infocomp2012_tutorial_sec.pdf.
- [9] C.-P. Rückemann, F. Hülsmann, and M. Hofmeister, "Cognostics and Applications for Knowledge Resources and Big Data Computing," KiM Summit, November 9, 2015, Knowledge in Motion, "Unabhängiges Deutsches Institut für Multi-disziplinäre Forschung (DIMF)", Hannover, Germany, 2015.
- [10] J. W. Tukey and P. A. Tukey, "Computer Graphics and Exploratory Data Analysis: An Introduction," in Proceedings of the Sixth Annual Conference and Exposition of the National Computer Graphics Association, Dallas, Texas, USA, April 14–18, 1985. Fairfax, VA: National

Computer Graphics Association, 1985, pp. 772–785, reprinted in W. S. Cleveland (ed.), The Collected Works of John W. Tukey, Vol 5 (pp. 419–436), New York: Chapman & Hall.

- [11] W. S. Cleveland, Dynamic Graphics for Statistics. Wadsworth & Brooks/Cole, Statistic/Probability Series, 1988, McGill, M. E. (ed.), ISBN: 0-534-09144-X.
- [12] T. L. Nyerges and D. M. Mark, Cognitive aspects of human-computer interaction for geographic information systems. Kluwer Academic Publishers, 1995.
- [13] M. F. Goodchild, Modelling error in objects and fields. Taylor & Francis, London, 1989, in: The Accuracy of Spatial Databases, Goodchild, M. F. and Gopal, S. (eds.), ISBN: 0-85066-847-6.
- [14] B. Tversky, "Cognitive Maps, Cognitive Collages, and Spatial Mental Models," Lecture Notes in Computer Science, vol. 716, 1993, pp. 14– 24, Plenum Press, New York.
- [15] D. R. Montello, "Scale and Multiple Psychologies of Space," in Spatial Information Theory: a theoretical basis for GIS, ser. Lecture Notes in Computer Science, A. Frank and C. I., Eds., no. 716, COSIT'93. Berlin: Springer Verlag, 1993, pp. 312–321.
- [16] G. Edwards, "The Voronoi Model and Cultural Space: Applications to the Social Sciences and Humanities," in Spatial Information Theory: A Theoretical Basis for GIS, ser. Lecture Notes in Computer Science, A. U. Frank and I. Campari, Eds., vol. 716. Berlin: Springer Verlag, 1993, pp. 202–214.
- [17] G. Edwards, "Geocognostics A New Paradigm for Spatial Information?" in Proceedings of the AAAI Spring Symposium 1996, 1996.
- [18] S. Hirtle, "Towards a Cognitive GIS, Proceedings of the Advanced Geographic Data Modelling Workshop," Journal of the Netherlands Geodetic Commission, vol. 40, 1994, pp. 217–227.
- [19] C.-P. Rückemann, "Dynamical Parallel Applications on Distributed and HPC Systems," International Journal on Advances in Software, vol. 2, no. 2, 2009, ISSN: 1942-2628, URL: http://www.iariajournals. org/software/, [accessed: 2009-11-16].
- [20] C.-P. Rückemann, "Integrating Future High End Computing and Information Systems Using a Collaboration Framework Respecting Implementation, Legal Issues, and Security," International Journal on Advances in Security, vol. 3, no. 3&4, 2010, pp. 91–103, savola, R., (ed.), pages 91–103, ISSN: 1942-2636, LCCN: 2008212457 (Library of Congress), URL: http://www.iariajournals.org/security/sec_v3_n34_ 2010_paged.pdf [accessed: 2015-06-07].
- C.-P. Rückemann, Integrated Information and Computing Systems [21] for Advanced Cognition with Natural Sciences. Premier Reference Source, Information Science Reference, IGI Global, 701 E. Chocolate Avenue, Suite 200, Hershey PA 17033-1240, USA, Oct. 2012, pp. 1-26, chapter I, in: Rückemann, C.-P. (ed.), Integrated Information and Computing Systems for Natural, Spatial, and Social Sciences, 543 (xxiv + 519) pages, 21 chapters, ill., ISBN-13: 978-1-4666-2190-9 (hardcover), EISBN: 978-1-4666-2191-6 (e-book), ISBN: 978-1-4666-2192-3 (print & perpetual access), DOI: 10.4018/978-1-4666-2190-9, LCCN: 2012019278 (Library of Congress), OCLC: 798809710. URL: http://www.igi-global.com/chapter/integratedinformation-computing-systems-advanced/70601 [accessed: 2015http://dx.doi.org/10.4018/978-1-4666-2190-9.ch001 08-02], DOI: [accessed: 2015-08-02].
- [22] C.-P. Rückemann, "From Multi-disciplinary Knowledge Objects to Universal Knowledge Dimensions: Creating Computational Views," International Journal On Advances in Intelligent Systems, vol. 7, no. 3&4, 2014, pp. 385–401, Bodendorf, F., (ed.), ISSN: 1942-2679, LCCN: 2008212456 (Library of Congress), URL: http://www.iariajournals.org/ intelligent_systems/intsys_v7_n34_2014_paged.pdf [accessed: 2015-06-07].
- [23] "Wissensarbeiter erfinden das Rad immer wieder neu," library essentials, LE_Informationsdienst, Mai/Juni 2015, 2015, pp. 30–32, ISSN: 2194-0126, URL: http://www.libess.de [accessed: 2015-06-07].
- [24] "The Knowledge Rework Report," 2015, December 2014, URL: http://

pages.coveo.com/KM-RESEARCH-SURVEY-RESULTS-LP.html [accessed: 2015-06-07].

- [25] "Industrie 4.0: Realität oder Wunschvorstellung?" library essentials, LE_Informationsdienst, Mai/Juni 2015, 2015, pp. 27–29, ISSN: 2194-0126, URL: http://www.libess.de [accessed: 2015-06-07].
- [26] "Jedem dritten Produktionsbetrieb ist Industrie 4.0 kein Begriff," 2015, Pressemitteilung vom 07. April 2015, URL: http://www.bitkom.org/de/ presse/8477_81970.aspx [accessed: 2015-06-07].
- [27] "4 von 10 Unternehmen nutzen Industrie 4.0-Anwendungen," 2015, Pressemitteilung vom 13. April 2015, URL: http://www.bitkom.org/de/ presse/8477_82018.aspx [accessed: 2015-06-07].
- [28] M. Rüßmann, M. Lorenz, P. Gerbert, M. Waldner, J. Justus, P. Engel, and M. Harnisch, "Industry 4.0: The Future of Productivity and Growth in Manufacturing Industries," 2015, Boston Consulting Group - BCG, April 2015, URL: http://www.bcgperspectives.com/Images/Industry_ 40_Future_of_Productivity_April_2015_tcm80-185183.pdf [accessed: 2015-06-07].
- [29] "Neue Bevölkerungsvorausberechnung für Deutschland bis 2060," 2015, Pressemitteilung Nr. 153 vom 28.04.2015, URL: https://www.destatis.de/DE/PresseService/Presse/Pressemitteilungen/ 2015/04/PD15_153_12421.html [accessed: 2015-06-07].
- [30] "Die Roboter kommen: Folgen der Automatisierung für den deutschen Arbeitsmarkt," 2015, Studie vom 30. April 2015, URL: https://www.ing-diba.de/imperia/md/content/pw/content/ueber_uns/ presse/pdf/ing_diba_economic_research_die_roboter_kommen.pdf [accessed: 2015-06-07].
- [31] H. Lau, "Four Ways Your Data is Lying to You," Datanami, 2015, June 8, 2015, URL: http://www.datanami.com/2015/06/08/four-waysyour-data-is-lying-to-you/ [accessed: 2015-06-16].
- [32] A. Woodie, "Big Data's Dirty Little Secret," Datanami, 2015, July 2, 2015, URL: http://www.datanami.com/2015/07/02/big-datas-dirtylittle-secret/ [accessed: 2015-07-05].
- [33] A. Gray, "Big Data's Small Lie The Limitation of Sampling and Approximation in Big Data Analysis," Datanami, 2015, July 20, 2015, URL: http://www.datanami.com/2015/07/20/big-datas-smalllie-the-limitation-of-sampling-and-approximation-in-big-data-analysis [accessed: 2015-07-26].
- [34] C. Lukasik, "Big Data, Big Misnomer," Datanami, 2015, July 22, 2015, URL: http://www.datanami.com/2015/07/22/big-data-bigmisnomer [accessed: 2015-07-26].
- [35] B. Gersbeck-Schierholz and C.-P. Rückemann, "All-in Folders: Challenges and Benefits," KiMrise, Knowledge in Motion, June 20, 2015, Circle Summit Workgroup Meeting, "Unabhängiges Deutsches Institut für Multi-disziplinäre Forschung (DIMF)", Düsseldorf, Germany, 2015.
- [36] "LX-Project," 2015, URL: http://www.user.uni-hannover.de/cpr/x/ rprojs/en/#LX (Information) [accessed: 2015-04-12].
- [37] E. Dodsworth and L. W. Laliberte, Eds., Discovering and using historical geographic resources on the Web: A practical guide for Librarians. Lanham: Rowman and Littlefield, 2014, ISBN: 0-8108-914-1.
- [38] S. A. Keller and R. Schneider, Eds., Wissensorganisation und -repräsentation mit digitalen Technologien. Walter de Gruyter GmbH, 2014, Bibliotheks- und Informationspraxis, ISSN: 0179-0986, Band 55, ISBN: 3-11-031270-0.
- [39] E. W. De Luca and I. Dahlberg, "Die Multilingual Lexical Linked Data Cloud: Eine mögliche Zugangsoptimierung?" Information Wissenschaft & Praxis, vol. 65, no. 4–5, 2014, pp. 279–287, Deutsche Gesellschaft für Information und Wissen e.V. (DGI), Ed., De Gruyter Saur, ISSN: 1434-4653, e-ISSN: 1619-4292, DOI: 10.1515/iwp-2014-0040, (title in English: The Multilingual Lexical Linked Data Cloud: A possible semantic-based access to the Web?).
- [40] L. Schumann and W. G. Stock, "Ein umfassendes ganzheitliches Modell für Evaluation und Akzeptanzanalysen von Informationsdiensten: Das Information Service Evaluation (ISE) Modell," Information Wissenschaft & Praxis, vol. 65, no. 4–5, 2014, pp. 239–246, Deutsche

Gesellschaft für Information und Wissen e.V. (DGI), Ed., De Gruyter Saur, ISSN: 1434-4653, e-ISSN: 1619-4292, DOI: 10.1515/iwp-2014-0043, (title in English: A comprehensive holistic model for evaluation and acceptance analyses of information services: The Information Service Evaluation (ISE) model).

- [41] G. Isaew and A. Roganow, "Qualitätssteuerung von Informationssystemen: Theoretisch-methodologische Grundlagen," Information Wissenschaft & Praxis, vol. 65, no. 4–5, 2014, pp. 271–278, Deutsche Gesellschaft für Information und Wissen e.V. (DGI), Ed., De Gruyter Saur, ISSN: 1434-4653, e-ISSN: 1619-4292 DOI: 10.1515/iwp-2014-0044, (title in English: Quality Management of Information Systems: Theoretical and methodological basics).
- [42] "Geo Exploration and Information (GEXI)," 1996, 1999, 2010, 2015, URL: http://www.user.uni-hannover.de/cpr/x/rprojs/en/index.html# GEXI [accessed: 2015-04-12].
- [43] C.-P. Rückemann, "Active Map Software," 2001, 2005, 2012, URL: http://wwwmath.uni-muenster.de/cs/u/ruckema (original permalink) [accessed: 2012-01-01], URL: http://www.user.uni-hannover.de/cpr/x/ publ/2001/dissertation/wwwmath.uni-muenster.de/cs/u/ruckema (information, data, abstract) [accessed: 2015-04-18].
- [44] C.-P. Rückemann, "Knowledge Integration for Scientific Classification and Computation," in The Fourth Symposium on Advanced Computation and Information in Natural and Applied Sciences, Proceedings of The 12th International Conference of Numerical Analysis and Applied Mathematics (ICNAAM), September 22–28, 2014, Rhodes, Greece, Proceedings of the American Institute of Physics (AIP), AIP Conference Proceedings, no. 1. AIP Press, American Institute of Physics, Melville, New York, USA, 2014, ISSN: 0094-243X, ISBN-13: 978-0-7354-1287-3, DOI: 10.1063/1.4912358.
- [45] C.-P. Rückemann, "Long-term Sustainable Knowledge Classification with Scientific Computing: The Multi-disciplinary View on Natural Sciences and Humanities," International Journal on Advances in Software, vol. 7, no. 1&2, 2014, pp. 302–317, ISSN: 1942-2628, LCCN: 2008212462 (Library of Congress), URL: http://www.iariajournals.org/ software/soft_v7_n12_2014_paged.pdf [accessed: 2015-06-07].
- [46] "GMT Generic Mapping Tools," 2015, URL: http://imina.soest.hawaii. edu/gmt [accessed: 2015-04-12].
- [47] C.-P. Rückemann, "High End Computing Using Advanced Archaeology and Geoscience Objects," International Journal On Advances in Intelligent Systems, vol. 6, no. 3&4, 2013, pp. 235– 255, ISSN: 1942-2679, LCCN: 2008212456 (Library of Congress), URL: http://www.iariajournals.org/intelligent_systems/intsys_v6_n34_ 2013_paged.pdf [accessed: 2015-04-12].
- [48] C.-P. Rückemann, "Enabling Dynamical Use of Integrated Systems and Scientific Supercomputing Resources for Archaeological Information Systems," in Proc. INFOCOMP 2012, Oct. 21– 26, 2012, Venice, Italy, 2012, pp. 36–41, ISBN: 978-1-61208-226-4, URL: http://www.thinkmind.org/download.php?articleid=infocomp_ 2012_3_10_10012 [accessed: 2015-04-12].
- [49] "Multilingual Universal Decimal Classification Summary," 2012, UDC Consortium, 2012, Web resource, v. 1.1. The Hague: UDC Consortium (UDCC Publication No. 088), URL: http://www.udcc.org/udcsummary/ php/index.php [accessed: 2015-04-12].
- [50] F. Cremer, C. Engelhardt, and H. Neuroth, "Embedded Data Manager – Integriertes Forschungsdatenmanagement: Praxis, Perspektiven und Potentiale," Bibliothek: Forschung und Praxis, vol. 39, no. 1 (April), 2015, pp. 13–31, ISSN: 0341-4183, DOI: http://dx.doi.org/10.1515/bfp-2015-0006 [accessed: 2015-05-10].
- [51] "Museum-Digital," 2015, URL: http://www.museum-digital.de/ [accessed: 2015-05-17].
- [52] G. Wurzer, K. Kowarik, and H. Reschreiter, Eds., Agent-based Modeling and Simulation in Archaeology. Springer, 2015, ISBN: 978-3-319-00008-4, URL: http://www.springer.com/978-3-319-00007-7 [accessed: 2015-05-17].
- [53] X. Rubio-Campillo, Large Simulations and Small Societies: High Per-

formance Computing for Archaeological Simulations. Springer, 2015, in: Agent-based Modeling and Simulation in Archaeology, Wurzer, G., Kowarik, K., and Reschreiter, H. (eds.), ISBN: 978-3-319-00008-4, URL: http://www.springer.com/978-3-319-00007-7 [accessed: 2015-05-17].

- [54] "Geschlossene Systeme als Bedrohung für das freie Web," library essentials, LE_Informationsdienst, Mai/Juni 2015, 2015, pp. 24–26, ISSN: 2194-0126, URL: http://www.libess.de [accessed: 2015-06-07].
- [55] J. Boie, "Schleichender Tod der Webseite," 2015, in: Süddeutsche Zeitung Online, 28. April 2015, URL: http://www.sueddeutsche. de/digital/ende-des-world-wide-web-das-netz-wird-neu-geknuepft-1.2454673 [accessed: 2015-06-07].
- [56] "The Domain Name Industry Brief," 2015, Vol. 12, No. 1, March 2015, URL: http://www.verisigninc.com/assets/domain-name-reportmarch2015.pdf [accessed: 2015-06-07].
- [57] T. Dunning, "The Role of Bias In Big Data: A Slippery Slope," Datanami, 2015, August 3, 2015, URL: http://www.datanami.com/2015/ 08/03/the-role-of-bias-in-big-data-a-slippery-slope/ [accessed: 2015-08-09].
- [58] T. Trader, "Reading List: Fault Tolerance Techniques for HPC," HPCwire, 2015, August 6, 2015, URL: http://www.hpcwire.com/2015/ 08/06/reading-list-fault-tolerance-techniques-for-hpc/ [accessed: 2015-08-15].
- [59] B. Gersbeck-Schierholz and C.-P. Rückemann, "Coping with Failures in Information and Applied Knowledge," KiMrise, Knowledge in Motion, June 20, 2015, Circle Summit Workgroup Meeting, "Unabhängiges Deutsches Institut für Multi-disziplinäre Forschung (DIMF)", Brussels-Welckenrath, Belgium, 2015.
- [60] C.-P. Rückemann, "Archaeological and Geoscientific Objects used with Integrated Systems and Scientific Supercomputing Resources," International Journal on Advances in Systems and Measurements, vol. 6, no. 1&2, 2013, pp. 200–213, ISSN: 1942-261x, LCCN: 2008212470 (Library of Congress), URL: http://www.thinkmind.org/download.php? articleid=sysmea_v6_n12_2013_15 [accessed: 2015-04-12].
- [61] "Tcl Developer Site," 2015, URL: http://dev.scriptics.com/ [accessed: 2015-04-12].
- [62] "Open-MPI," 2015, URL: http://www.open-mpi.org [accessed: 2015-04-12].
- [63] R. Suchenwirth, "Tclworld, Version 0.6, Konstanz, 2002-01-31," 2002, URL: http://wiki.tcl.tk/_repo/wiki_images/tm06.zip [accessed: 2015-04-12].
- [64] R. Suchenwirth, "dbapi A simple database API," 2002, URL: http: //wiki.tcl.tk/_repo/wiki_images/tm06.zip [accessed: 2015-04-12].
- [65] R. Suchenwirth, "dbgui.tcl A little database GUI," 2002, URL: http: //wiki.tcl.tk/_repo/wiki_images/tm06.zip [accessed: 2015-04-12].
- [66] W. Gaul and M. Schader, Eds., Classification As a Tool of Research. North-Holland, Amsterdam, 1986, Proceedings, Annual Meeting of the Classification Society, (Proceedings der Fachtagung der Gesellschaft für Klassifikation), ISBN-13: 978-0444879806, ISBN-10: 0-444-87980-3, Hardcover, XIII, 502 p., May 1, 1986.
- [67] P. Cousson, "UDC as a non-disciplinary classification system for a high-school library," in Proceedings UDC Seminar 2009, Classification at a Crossroads: Multiple Directions to Usability, 1992, pp. 243–252, URL: http://www.academia.edu/1022257/UDC_as_a_non-disciplinary_ classification_system_for_a_high-school_library [accessed: 2015-04-12].
- [68] A. Adewale, "Universal Decimal Classification (UDC): A Most For All Libraries," library 2.0, the future of libraries in the digital age, 2014, URL: http://www.library20.com/forum/topics/universal-decimalclassification-udc-a-most-for-all-libraries [accessed: 2015-04-12].
- [69] J. Templin and L. Bradshaw, "Measuring the Reliability of Diagnostic Classification Model Examinee Estimates," Journal of Classification, vol. 30, no. 2, 2013, pp. 251–275, Heiser, W. J. (ed.), ISSN: 0176-4268

(print), ISSN: 1432-1343 (electronic), URL: http://dx.doi.org/10.1007/s00357-013-9129-4 [accessed: 2015-04-12].

- [70] P. Cerchiello and P. Giudici, "Non parametric statistical models for online text classification," Advances in Data Analysis and Classification – Theory, Methods, and Applications in Data Science, vol. 6, no. 4, 2012, pp. 277–288, special issue on "Data analysis and classification in marketing" Baier, D. and Decker, R. (guest eds.) ISSN: 1862-5347 (print), ISSN: 1862-5355 (electronic).
- [71] F. Heel, "Abbildungen zwischen der Dewey-Dezimalklassifikation (DDC), der Regensburger Verbundklassifikation (RVK) und der Schlagwortnormdatei (SWD) für die Recherche in heterogen erschlossenen Datenbeständen – Möglichkeiten und Problembereiche," Bachelorarbeit im Studiengang Bibliotheks- und Informationsmanagement, Fakultät Information und Kommunikation, Hochschule der Medien Stuttgart, 2007, Bearbeitungszeitraum: 1. Juni 2007 bis 31. August 2007, Stuttgart, August 2007, URL: http://opus.bsz-bw.de/hdms/volltexte/2009/665/pdf/ BA_Fabian_Heel.pdf [accessed: 2015-04-12].
- [72] "Physics and Astronomy Classification Scheme, PACS 2010 Regular Edition," 2010, American Institute of Physics (AIP), URL: http://www. aip.org/pacs [accessed: 2015-04-12].
- [73] "Mathematics Subject Classification (MSC2010)," 2010, URL: http:// msc2010.org [accessed: 2015-04-12].
- [74] R. S. Sasscer, U.S. Geological Survey library classification system. U.S. G.P.O., 1992, USGS Bulletin: 2010.
- [75] "North American Industry Classification System (NAICS), Concordances," 2014, URL: https://www.census.gov/eos/www/ naics/concordances/concordances.html [accessed: 2015-04-12].
- [76] F. Hülsmann and C.-P. Rückemann, "Intermediate Classification and Concordances in Practice," KiMrise, Knowledge in Motion Meeting, January 9, 2015, Knowledge in Motion, "Unabhängiges Deutsches Institut für Multi-disziplinäre Forschung (DIMF)", Hannover, Germany, 2015.
- [77] "Universal Decimal Classification Consortium (UDCC)," 2015, URL: http://www.udcc.org [accessed: 2015-04-12].
- [78] A. Slavic, "UDC libraries in the world 2012 study," universaldecimalclassification.blogspot.de, 2012, Monday, 20 August 2012, URL: http://universaldecimalclassification.blogspot.de/2012/08/ udc-libraries-in-world-2012-study.html [accessed: 2015-04-12].
- [79] "Creative Commons Attribution Share Alike 3.0 license," 2012, URL: http://creativecommons.org/licenses/by-sa/3.0/ [accessed: 2015-04-12].
- [80] G. Leopold, "Can Containers Contain?" Remains Top Security Issue," Enterprisetech, 2015, July 29, 2015, URL: http://www.enterprisetech.com/2015/07/29/can-containers-containremains-top-security-issue/ [accessed: 2015-08-02].
- [81] G. Leopold, "Goldman Sachs Brokers Container Spec Deal," Enterprisetech, 2015, June 22, 2015, URL: http://www.enterprisetech. com/2015/06/22/goldman-sachs-brokers-container-spec-deal/ [accessed: 2015-08-02].
- [82] K. Kincade, "NERSC's 'Shifter' Makes Container-based HPC a Breeze," HPCwire, 2015, August 7, 2015, URL: http://www.hpcwire.com/2015/08/07/nerscs-shifter-makes-containerbased-hpc-a-breeze/ [accessed: 2015-08-15].